### THESIS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

# A comprehensive approach to building-stock modelling

Assessing the impact of renovating urban housing stocks

# MAGNUS ÖSTERBRING

Department of Architecture and Civil Engineering CHALMERS UNIVERSITY OF TECHNOLOGY Gothenburg, 2019 A comprehensive approach to building-stock modelling Assessing the impact of renovating urban housing stocks MAGNUS ÖSTERBRING ISBN: 978-91-7905-179-2

© MAGNUS ÖSTERBRING 2019

Doktorsavhandlingar vid Chalmers tekniska högskola Ny serie nr: 4646 ISSN 0346-718X

Department of Architecture and Civil Engineering Chalmers University of technology SE-412 96 Gothenburg Sweden Telephone +46 (0)31-772 1000

Printed by Chalmers Reproservice Gothenburg, Sweden 2019 A comprehensive approach to building-stock modelling Assessing the impact of renovating urban housing stocks Magnus Österbring Department of Architecture and Civil Engineering Division of Building Technology Chalmers University of Technology

## Abstract

The existing building stock provide a possibility for cost-efficient energy efficiency measures and related reductions in greenhouse-gas emissions. As the rate of renewal in the buildingstock is low, energy efficiency measures need to be applied when renovation is being done in order to reach climate goals. To increase the renovation rate and realise the potential for substantial reductions in energy use, several research and demonstration projects have been carried out on both a European and Swedish level. In order to evaluate the current state and renovation potential of the existing building stock on an urban level, a local approach is needed to understand challenges and possibilities associated with the transformation of the building-stock. To quantify the potential for reducing energy use and greenhouse-gas emissions, building-stock modelling is commonly used. However, these models are often based on using representative buildings and scaling factors. With increased spatial resolution, building descriptions based on representative buildings lose accuracy and as a result, stakeholders operating at a planning or policy level are commonly targeted. This study proposes a building-specific stock description where each building is treated individually to differentiate the renovation potential within the building-stock. For this purpose, available databases containing building-specific information has been gathered and processed for the multi-family building stock of the City of Gothenburg. The available data is used to create a building-specific stock description and renovation measures are modelled using a bottom-up engineering method and evaluated regarding energy use, environmental impact and costeffectiveness. This thesis with appended papers shows that available data sources can be used to describe the characteristics of the stock on a building level and model the effect of renovation on energy use, environmental impact and cost-effectiveness in order to provide detailed information to policy makers, planners and property owners.

Keywords: Building-stock modelling, energy performance certificate, GIS, LCA, costeffectiveness, building valuation, multi-family buildings, energy, renovation, refurbishment

# Acknowledgements

This thesis summarizes the studies at the division of Building Technology in the research group of Sustainable Building, Chalmers University of Technology. The work has been financed by the Swedish Energy Agency, the Climate-KIC, Chalmers university of technology and NCC AB. This study has been made possible with the help of data supplied by the City planning office of the City of Gothenburg, Swedish Tax Agency as well as Riksbyggen.

I would like to thank my colleagues at ACE as well as my colleagues at NCC. I would especially like to thank my supervisors Holger Wallbaum, Liane Thuvander and Christina Claeson-Jonsson for their support and feedback throughout this process. I would also like to thank Filip Johnsson, Érika Mata, Mikael Mangold, Claudio Nägeli and Clara Camarasa for their collaboration in this work. Lastly, I would like to thank my family, friends and my sambo.

# **List of Publications**

This thesis is based on the following four peer-reviewed journal papers:

- Österbring, M., Mata, É., Thuvander, L., Mangold, M., Johnsson, F., & Wallbaum, H. (2016). A differentiated description of building-stocks for a georeferenced urban bottom-up building-stock model. Energy and Buildings, 120, 78-84.
- II. Österbring, M., Thuvander, L., Mata, É., & Wallbaum, H. (2018). Stakeholder Specific Multi-Scale Spatial Representation of Urban Building-Stocks. ISPRS International Journal of Geo-Information, 7(5), 173.
- III. Österbring, M., Camarasa, C., Nägeli, C., Thuvander, L., & Wallbaum, H. (2019).
   Prioritizing deep renovation for housing portfolios. *Energy and Buildings*, 109361.
- IV. Österbring, M., Mata, É., Thuvander, L., & Wallbaum, H. (2019). Explorative lifecycle assessment of renovating existing urban housing-stocks. Building and Environment, 106391.

The following publications are not included in the thesis but listed here for further reading.

- V. Österbring, M., Mata, É., Johnsson, F., & Wallbaum, H. (2014). A methodology for spatial modelling of energy and resource use of buildings in urbanized areas. In World Sustainable Building Conference WSB14 Barcelona.
- VI. Mangold, M., Österbring, M., & Wallbaum, H. (2015). A review of Swedish residential building stock research. International Journal of Environmental Sustainability.
- VII. Mangold, M., Österbring, M., & Wallbaum, H. (2015). Handling data uncertainties when using Swedish energy performance certificate data to describe energy usage in the building stock. Energy and Buildings, 102, 328-336.
- VIII. Thuvander, L., Österbring, M., Mangold, M., Mata, E., Wallbaum, H., & Johnsson, F. (2015). Spatial exploration of the refurbishment dynamics of urban housing stocks. In Computers in Urban Planning and Urban Management CUPUM Cambridge.
  - IX. Mangold, M., Österbring, M., Wallbaum, H., Thuvander, L., & Femenias, P. (2016) Socio-economic impact of renovation and energy retrofitting of the Gothenburg building stock. Energy and Buildings, 123, 41-49.
  - X. Österbring, M., Thuvander, L., Mata, É., & Wallbaum, H. (2017). Renovation Needs and Potential for Improved Energy Performance Depending on Ownership – A Location Based Study of Multi-Family Building Stocks in an Urban Context. In World Sustainable Building Conference WSBE17 Hong-Kong.
  - Mangold, M., Österbring, M., Overland, C., Johansson, T., & Wallbaum, H. (2018).
     Building Ownership, Renovation Investments, and Energy Performance—A Study of Multi-Family Dwellings in Gothenburg. Sustainability, 10(5), 1684.
- XII. Österbring, M., Rosado, L., Wallbaum, H., & Gontia, P. (2018). An Approach to Identify Resource Patterns on a Neighborhood Level. In Factor X (pp. 317-323). Springer, Cham.
- XIII. Gontia, P., Nägeli, C., Rosado, L., Kalmykova, Y., & Österbring, M. (2018). Material-intensity database of residential buildings: A case-study of Sweden in the international context. *Resources, Conservation and Recycling*, 130, 228-239.

- XIV. Mata, É., Wanemark, J., Österbring, M., Thuvander, L., Wallbaum, H. (2018). Decision-Making in Building Retrofitting: Lessons from Dynamic Modelling of Scenarios for Gothenburg City, in 5th European Conference on Behaviour and Energy Efficiency, 2018.
- XV. Nägeli, C., Farahani, A., Österbring, M., Dalenbäck, J. O., & Wallbaum, H. (2019). A service-life cycle approach to maintenance and energy retrofit planning for building portfolios. Building and Environment, 106212.
- XVI. Eriksson, S., Waldenström, L., Tillberg, M., Österbring, M., & Sasic Kalagasidis, A. (2019). Numerical Simulations and Empirical Data for the Evaluation of Daylight Factors in Existing Buildings in Sweden. Energies, 12(11), 2200.

## Acronyms and nomenclature

#### Acronyms

- BSM Building-stock modelling
- DHW Domestic hot-water
- EAC Equivalent annual cost
- EEM Energy efficiency measure
- EPC Energy performance certificates
- GIS Geographic information systems
- HFA Heated floor area
- $LCA-Life\mbox{-cycle}\ assessment$
- MFB Multi-family building
- SH Space heating

#### Definitions

Deep renovation – A comprehensive renovation of a building aiming at considerable reductions in energy use.

Energy conservation measure – A measure that reduces energy demand and may impact the function, i.e. reducing indoor set-point temperature.

Energy cost saving - The monetary savings from reducing energy use.

Energy efficiency measures – A measure that reduces energy demand without impacting the function, i.e. improving the thermal insulation of a building component. Energy efficiency measures are a subsection of energy conservation measures.

Final energy – Energy delivered to the building for heating, domestic hot-water and auxiliary electricity use. It does not include household electricity use.

Multi-family building – The Swedish definition of a multi-family building is a building with three or more apartments.

Renovation – A change to a building or a building component in order to restore or improve the original function.

# Contents

ABSTR	ACT III
ACKNO	WLEDGEMENTSV
LIST OF	PUBLICATIONS
ACRON	IYMS AND NOMENCLATUREIX
CONTE	NTSXI
1. II	NTRODUCTION1
1.1	AIM
1.2	SCOPE
1.3	METHODOLOGY
1.4	STRUCTURE OF THE THESIS
2. B	UILDING-STOCK MODELLING7
2.1	A BRIEF HISTORY OF BUILDING-STOCK MODELLING
2.2	BUILDING-STOCK DESCRIPTION USING REPRESENTATIVE BUILDINGS
2.3	BUILDING-SPECIFIC STOCK DESCRIPTION
2.4	ENERGY MODELLING OF BUILDING STOCKS
2.5	ECONOMIC MODELLING OF BUILDING STOCKS
2.6	ENVIRONMENTAL MODELLING OF BUILDING STOCKS
2.7	VISUALIZATION AND COMMUNICATION
2.8	TOWARDS COMPREHENSIVE BUILDING-STOCK MODELLING
3. B	UILDING-STOCK INFORMATION15
3.1	DATA SOURCES
3.2	COMBINING DATASETS
4. B	UILDING-SPECIFIC STOCK DESCRIPTION AND MODELLING FRAMEWORK
4.1	DETERMINING THE U-VALUE
4.2	MODELLING OF ENERGY USE, ENVIRONMENTAL IMPACT AND COST-EFFECTIVENESS
4.3	Assessing changes to the building-stock
4.4	SPATIAL VISUALISATION
5. R	ESULTS
5.1	CURRENT STATE OF THE STOCK
5.2	RENOVATING THE EXISTING STOCK
6. C	DISCUSSION AND CONCLUSIONS
7. F	UTURE RESEARCH
8. R	EFERENCES

#### 1. Introduction

In Europe, buildings account for about 40 % of the final energy use and 36 % of CO<sub>2</sub>emissions [1] and provide an opportunity for cost-efficient energy efficiency measures (EEM) [2]. The European Energy Performance of Buildings Directive defines efficiency standards for both new and existing buildings targeting these efficiency opportunities [3], [4]. On a national level, the Swedish government has set policy which strives for substantial reductions in energy use by 2020 and 2050 [5], [6]. As a result of the EU energy efficiency directive [4], energy performance certificates (EPC) were introduced in Sweden in 2006 in order to encourage energy efficiency in buildings. On a local level, more ambitious targets on energy savings have voluntarily been adopted by cities and municipalities. The city of Gothenburg has implemented such targets and aims to reduce energy consumption in residential buildings by 30 % by 2020 compared to 1995 levels [7]. For developed countries it is estimated that most of the buildings that will be in use in 2050 have already been built [8]. Meanwhile, the renewal rate of the Swedish residential stock is only 0.6% [9] which implies a need for EEM in the existing stock if these targets are to be met.

According to Statistics Sweden [10], the Swedish residential building stock consists of roughly 4.9 million apartments of which 2.5 million are found in the multi-family building (MFB) stock. The MFB stock is old with 75 % of apartments having been built before 1980 and 51 % of apartments being built between 1951 and 1980. Many of these apartments were built as a result of a governmental programme in 1965, the million homes programme, aiming at one million new apartments in the coming decade [11]. During this time period, roughly 700 000 apartments in MFBs were constructed. Furthermore, many of these buildings have not been renovated and there have been a few attempts to quantify the renovation need. In 2011, a survey conducted with property owners came to the conclusion that roughly 75 % of the MFBs from 1961-1975 were due for renewal, with 320 000 apartments needing thorough renovation [12]. A similar conclusion was made by a study in 2013, estimating 614 000 apartments needing renovation in the near future of which 471 000 needing major renovation [13]. A follow-up study was done in 2018 by accounting for major renovation activities since 2013 and concluded that there are still 213 000 apartments in need of major renovation [14].

By European standards, the energy performance of the Swedish housing stock is high and in combination with comparatively low greenhouse-gas emissions due to district heating, the average greenhouse-gas emissions for the residential sector is about a third of that in the EU27 [1]. The average energy use for space heating (SH) and domestic hot-water (DHW) for the 205 million m<sup>2</sup> of heated floor area (HFA) in the Swedish MFB stock has been roughly 27 TWh over the last decade with over 90 % being covered by district heating [15]. The average energy performance in the MFB stock for SH and DHW is 138 kWh/m<sup>2</sup>. The energy performance is somewhat lower in older parts of the stock, 147 kWh/m<sup>2</sup> for buildings built before 1961, and 140 kWh/m<sup>2</sup> for buildings built between 1961 and 80 [16]. The yearly rate of renovation has been estimated at around 1 % [17] and as no discernible reductions in total energy use in the stock has occurred in the last decade, it is likely that any energy demand reductions achieved in the existing stock has been at least partly offset by new construction. It

has also been noted that the main barrier for increasing the renovation rate is profitability and due to the widespread use of district heating, deep renovation only providing marginal greenhouse-gas reductions [17].

To increase the renovation rate and realise the potential for substantial reductions in energy use, several research and demonstration projects have been carried out on both a European and Swedish level. Over the past decade, there have been several EU funded research projects focusing on energy efficiency measures such as HERB (Holistic energy-efficient retrofitting of residential buildings), E2ReBuild (Industrialised energy efficient retrofitting of resident buildings in cold climates) and RETROKIT (Toolboxes for systemic retrofitting) to name a few. In Sweden, two methods have been developed and applied for deep renovation of residential [18] and commercial [19] buildings respectively. However, while these methods and case studies provide useful tools in assessing the cost-effectiveness of renovation for a specific building, they are not applicable to the building stock as a whole.

Building-stock modelling (BSM) has previously been used to assess the energy demand of the existing stock, prioritize what measures to apply and where they would be most effective. Much work has gone in to develop the energy simulation in BSM but little work has been done taking advantage of the building-specific information that is available. In order to evaluate renovation potential of the existing building stock on an urban level, a space and context specific approach is needed to understand challenges and possibilities associated with renovating the building-stock.

### 1.1 Aim

The aim of this study is to evaluate the potential for renovation by expanding on the methods used in building-stock modelling in order to account for environmental impact, cost-effectiveness, target a wider range of stakeholders with a particular focus on property owners and to visualize and communicate results using geographic information systems (GIS). In order to do so, the following research questions have been formulated based on the problem description and a literature study. How the research questions relate to the appended papers is shown in figure 1.

• Research question 1

What is the potential for a building-stock description where each building is treated individually?

• Research question 2

How can GIS be used with a building-specific stock description to visualize and communicate results for a wider range of stakeholders, including property owners?

• Research question 3

What are the environmental impacts of current renovation trends in an urban building-stock?

• Research question 4

How can the financial viability of (deep) renovation be assessed across a building portfolio?



Figure 1 - Research questions and related papers

### 1.2 Scope

The MFB stock in the City of Gothenburg is used as a case study. The challenge of renovating the urban housing-stock of the City of Gothenburg is evaluated using a building-stock description that enables modelling the effect of renovation activities regarding energy performance, environmental impact and financial viability for each individual building in the stock. As such, the scope of this thesis is limited to MFBs in a Swedish urban context.

### 1.3 Methodology

The methodology used in this thesis is based on BSM using a building-specific stock description. It consists of the following four steps: (1) data acquisition and processing, (2) developing a building-specific description of the stock, (3) modelling as well as (4) evaluation and visualization. These steps are briefly described below.

### Data acquisition and processing

Data has been gathered from national databases such as the property registry, the building registry, the national registry of energy performance certificates and spatially linked to a local 2.5D GIS model of the City of Gothenburg. The building-stock information available in Sweden and how it is processed to describe the characteristics of the MFB stock of Gothenburg is presented in more detail in chapter 3 and in paper I-IV.

## Creating a building-stock description

Based on available building-stock information, a description detailing the characteristics of each building in the stock is developed where aspects such as U-values, heating and ventilation system, surface areas and volume are determined. As the available data does not directly cover all aspects needed for modelling, estimations are made based on historic building regulations and architectural history books. A thorough description of the steps involved can be found in chapter 4 as well as in paper I, III and IV.

## Modelling

Depending on the research question, different modelling approaches have been used. Generally, modelling the energy use of the building-stock forms the foundation. In this work, two different approaches to modelling the energy performance of building-stocks has been used, ECCABS [20] in paper I and IV and the energy model described in [21] in paper III. In addition, a life-cycle assessment (LCA) has been conducted using SimaPro V8.0.5.13 based on environmental data from Ecoinvent and assessed using ReCIPe [22] V1.12 mid-point categories. Economic modelling has been conducted using equivalent annual cost (EAC) as well as the Swedish tax agency's model for assessing change in building value due to renovation activities. The methodology is expanded on in chapter 4.

### Visualization

GIS based visualizations have been used to investigate and showcase spatial patterns or the lack thereof. Visualizations have been done using ArcGIS using different levels of spatial aggregation in a first attempt to tailor results to a wider range of potential stakeholders.

## 1.4 Structure of the thesis

This thesis consists of a summary compilation and four appended papers. The summary consists of seven chapters containing the following:

- Chapter 1 presents the background for the thesis as well as aim, research questions and a brief methodological overview.
- Chapter 2 contains a comprehensive overview of the research field, further adding to the problem description, as well as a theoretical framework for BSM.

- Chapter 3 accounts for the data sources and attributes used in creating a building-specific stock description for the MFB stock in the City of Gothenburg.
- Chapter 4 details the methodology used to create a building-specific stock description based on available data, how changes to the building stock is implemented and evaluated as well as how visualizations have been done.
- Chapter 5 presents the main results of the work by showing the current state of the MFB stock and summarizes the results of assessing the impact of renovating the existing stock found in the appended papers.
- Chapter 6 contains a discussion and conclusion based on the work presented in the thesis.
- Chapter 7 describes potential future work that has been identified.

# 2. Building-stock modelling

Building-stock modelling (BSM) has been used over the past decades to model the current state of the building stock and to evaluate possible changes. Over time, the methodologies have become more advanced and the focus has expanded from energy use to a wider range of indicators such as embodied energy, greenhouse-gas emissions and environmental impact. In this chapter, the development of the BSM field is described and development potentials are identified, resulting in a comprehensive framework for bottom-up engineering-based building-stock modelling.

## 2.1 A brief history of building-stock modelling

BSM has traditionally been used to evaluate energy performance of building-stocks. There are many examples of BSM being used to evaluate the energy demand of the existing building-stock [23]–[28] and several in-depth review papers exists [29]–[32]. BSM typically has three or four distinct methodological steps, depending on whether the current state of the building-stock is to be assessed or if potential changes are to be evaluated. First, a building-stock description is developed to be used as input for modelling. Second, if changes to the building-stock are to be evaluated, a scenario is defined for the future development of the building-stock. Third, relevant parameter(s) are modelled, typically consisting of modelling the energy use and/or energy use reduction potential. Fourth, results are aggregated to a suitable spatial scale, compiled and presented through graphs, tables and map-based visualizations.

In order to model energy use of building-stocks, the overarching approach typically falls within one of two categories, either bottom-up or top-down. The top-down approach treats the building-stock as an energy sink where energy use is impacted due to changes on a macro level. Parameters commonly used are rate of new construction, renovation rate, population growth and costs relating to energy use and construction activities. This also means that energy use for specific end-uses such as SH or DHW is not directly observed as the level of detail is typically limited to the entire stock or housing sector. As such, the approach is not suitable for studying the effects of specific construction or renovation measures. Furthermore, as the variables used are based on historical data, technology development resulting in a shift from common construction practices cannot be accounted for. As the aim of this work is to allow for a more detailed assessment of renovation options, a top-down approach is not suitable.

Bottom-up models can be divided into two sub-groups; statistical models and bottom-up engineering models. Statistical models use aggregated data as input, which through regression methods account for specific end-uses based on the energy consumption of buildings. Bottom-up engineering models use a heat balance model to estimate the energy consumption for individual buildings. The buildings used as input in bottom-up models are defined by building properties such as geometry, U-values, climate data, indoor temperature and use of appliances. Thus, to apply a bottom-up engineering model requires detailed input data. Due to limited data availability and computational time-constraints the building stock is normally represented by sample buildings or archetype buildings, where it is assumed that similar buildings with regard to year of construction, use of the building, type of heating system can be represented by an average building. Sample buildings use detailed data for a selection of

buildings (e.g. as obtained from measurements or site inspection of individual buildings) combined with weighting factors for the sample buildings to reflect the entire building stock. Similarly, archetype buildings use representative theoretical buildings, often defined by construction year and the type or use of the building, to represent all buildings with similar characteristics to allow for assessment of the entire stock. These methods of developing a building description have been successfully used to calculate the potential for EEM in existing residential building-stocks on a national scale [33], [34] as well as on an urban scale [35]–[37].

Recent improvements in data availability have allowed greater focus on urban settings in BSM and include a spatial dimension by integrating geo-referenced data using geographical information systems (GIS) [38]–[40]. Using GIS in BSM has several advantages as it enables merging of data from different databases, it enables further analysis and communication by spatially differentiating and visualizing results, and finally it provides a solution for storing and exchanging data through interconnected urban models. The addition of a GIS component to BSM has been carried out to analyse energy policy scenarios in an urban context [41], to assess the urban heat island effect on energy demand [40] as well as to assess environmental impacts of building stocks and potential for EEM [39]. Further developments have been made by incorporating building specific data, most commonly taking advantage of 3D city models based on LIDAR data [42] or by analysing differences in digital terrain models and digital elevation models [43] as well as using building-specific data from EPC to better describe the technical characteristics of individual buildings [44]. This development allows for the possibility of visualizing and communicating results on a building level as well as allowing results to be aggregated arbitrarily to suit communication with different stakeholders.

### 2.2 Building-stock description using representative buildings

While the introduction of GIS in BSM has allowed an increase in spatial resolution and enabled focus on urban settings, using a description of the building-stock based on representative buildings has not been adapted to take full advantage of the improved potential for describing each building individually. Using representative buildings to model the stock can be problematic, typically so for older parts of the stock where renovations have been applied to a varying degree which may result in significant differences in the energy performance for the same type of buildings and the energy models become more advanced, the building-stock descriptions used as input for these models have seen little development and are still largely based on using representative buildings which are used to scale results to the desired level of output. Such descriptions loose accuracy with increased spatial resolution and commonly results are only presented at aggregate levels for districts or entire cities and not on a building level [31].

As a building stock description based on representative buildings limits accurate results to higher levels of aggregation, stakeholders operating at a planning or policy level are commonly targeted. A few exceptions can be found where other potential stakeholders have been identified. It has been suggested that construction companies can use EPC data to assess the size of the renovation market [47] and [48] points to the possibility to use results for

educational purposes. As such, the intended stakeholders for BSM can generally be divided into three broad categories: urban planners, energy planners and governmental bodies needing policy support. However, there is a lack of studies using BSM to target property owners and managers who are essential to the urban transformation process.

To support planners and policy makers, base-line models of the existing stock have been used to assess the current energy performance of cities and districts to highlight areas where interventions should be prioritized on both the energy demand side [49] and the supply side [50]. Similarly, to assess the technical potential of specific technologies, specific measures have been investigated to evaluate the potential to reach environmental targets at an urban level using EEM [51], the potential development of renewables [37] and potential expansion and optimal layout of district heating networks [52]. While these models tend to focus on the technical potential, other models have developed dynamic scenarios to describe the change of the building-stock over time. These scenarios range from assumptions on a fixed rate of technology implementation to agent-based models or other decision models based on economic [53] or socio-economic feasibility [54].

### 2.3 Building-specific stock description

A building-specific stock description, sometimes referred to as a building-by-building description is what has been developed in this thesis. It treats each building individually rather than relying on using representative buildings and scaling factors. To do so increases the already high demand on data availability as detailed information about each individual building in the stock is needed. However, if issues relating to data availability can be overcome a building-specific stock description has several advantages. First, it provides a potential for a higher degree of accuracy, especially in evaluating potential changes to the existing building-stock. While a representative description can capture the average impact of a renovation measure for a representative building, a building-specific description allows for a fuller picture where the distribution of the impact across the building-stock can be assessed. This will provide better understanding and decision support in cases where a representative description can indicate whether a renovation measure is profitable while the building-specific description will highlight for how many buildings that is the case and what the distribution is. Ideally, this will allow for prioritizing renovation measures for individual buildings within a larger property portfolio. In combination with building-level measured energy use for validation, results can be aggregated arbitrarily where models using representative buildings are typically tied to geographic boundaries and scales where measured energy use is available. As such, it allows for targeting stakeholders such as property owners and managers. Another use of detailed building-by-building stock information is for statistical analysis which has been used to study investments in renovation based on ownership [55], to evaluate the performance gap post renovation [56], [57] and to assess the renovation rate of non-profit housing [58].

### 2.4 Energy modelling of building stocks

To estimate the energy use of a building, all energy end-uses are typically considered. These end-uses can be broadly divided into SH, hot water use, lighting and appliances as well as auxiliary energy for building operations such as fans and pumps. In addition, there are interdependencies where appliance and interior lighting will affect the heating demand. The energy demand can then be supplied by different energy carriers such as the electricity grid, a district heating system or in-situ generation. Energy modelling of buildings aim to quantify energy demand for different end-uses based on a set of input parameters. The level of detail of input parameters varies depending on purpose, data availability and assumptions made. Depending on the model, the input parameters may be on a macro scale (renovation rate, energy price, inflation) or on a micro scale (U-values, technical systems, building geometry). Quantifying energy use of larger building-stocks and assessing changes to the stock due to renovation and new construction can support decision making regarding energy supply, renovation incentives, the building code and to develop pathways to reach environmental goals.

The energy modelling techniques used to calculate the energy performance of building-stocks have been detailed in several papers [20], [30]-[32]. A wide range of approaches have been used to model the energy demand of building-stocks. Of particular importance is the choice of spatial and temporal resolution of the heat-balance as it impacts computational time. The spatial resolution relates to the number of thermal zones used to model a buildings energy demand. Further complexity can be added if these thermal zones are interconnected. Using a single zone to model the energy demand of a building comes with several limitations. First, using a single thermal zone only allows for using a single temperature set-point which makes it less suitable for buildings with a mixed use. Second, it is not possible to account for a simultaneous heating and cooling load. This is typically problematic for buildings with large solar gains as this may cause a simultaneous heating and cooling load in different parts of the building. In the case of a single-zone model, the heating and cooling load would cancel each other out. Temporal resolution deals with the time-step used in the energy demand calculation, typically using a yearly, monthly, hourly or an adaptive time-step. In addition to impacting the computational time, an increase in temporal resolution also warrants input data with a higher resolution regarding outdoor climate conditions, user-behaviour, solar gains, thermal mass and control schedules. Many urban building energy models focus on domestic buildings as the spatial and temporal resolution of the energy calculation has a lower impact than for commercial buildings with complex heating, ventilation and air-conditioning systems. Engineering-based bottom-up models put high requirements on level of detail and available data regarding the technical aspects of buildings. However, user behaviour has not gathered the same attention. A recent review of user behaviour in urban energy models [59] shows that standardized deterministic space-based occupant behaviour is often used in conjunction with an archetype based description of the stock. The few cases where a more nuanced view on user behaviour is used was observed for single-use districts, either office or residential buildings. Another approach to handle the complexities of user behaviour in urban energy models is the use of synthetic populations [60]. This would allow for dynamic assessment of internal gains and loads resulting in a more detailed energy use profile. This would be beneficial when studying a local energy system to be able to assess interventions in the stock that could help reduce peak loads or to better assess greenhouse-gas emission reductions.

To calibrate and validate urban building energy models, measured energy use on a stock level is often used. This becomes problematic when models are based on a representative description of the building-stock and as it is not possible to separate the validity of the energy modelling from the representative description being used. Similarly, using aggregated measured energy use to calibrate a representative stock description is problematic as the energy use for the different representative buildings is unknown. One of the most commonly cited obstacles for handling uncertainty in BSM is the lack of disaggregated measured energy use data [29], [35], [42], [46], [61].

### 2.5 Economic modelling of building stocks

In addition to modelling the energy performance of renovation measures for building stocks, the cost and profitability of renovation measures are sometimes evaluated. This has been done by assessing energy saving measures for the Danish housing stock [62], cost and costeffectiveness of renovating the Swedish residential stock [20], cost of energy conservation and solar systems for existing multi-family buildings in Thessaloniki [37] and economic feasibility of deep renovation of the housing stock in Bologna [63]. The economic impact of EEM has been a subject of study in the past years [64]–[66]. Over the last decade, there have been several large EU funded projects focusing on EEM in the existing stock (see chapter 1). In most cases, these projects focused on assessing the viability and effect of EEM on individual buildings. In addition, there are several papers with a similar aim. The most profitable combination of insulation and glazing have been studied [67] and an optimization mode to define cost-effective measures in order to minimize energy use has been developed [68]. Moreover, several studies evaluated the economic viability and impact of EEM in the housing stock [69], [70], some of which used EPC for their assessments [71]–[74]. In general, economic modelling is typically done by evaluating the total cost of renovation in order to estimate market potential, the cost-effectiveness or profitability of measures to assess the techno-economical potential for renovating the building-stock or to assess the prospect of certain renovation measures. In addition, there are numerous case studies dealing with costeffectiveness of renovation on a building level and this has been the key focus of many EU funded research projects. Assessment ranges from pay-back time to life-cycle costing methodologies. However, as most assessment is carried out for a specific building or for representative buildings, the distribution of cost-effectiveness for the entire building stock is unknown. There is also potential in using building-stock modelling to investigate policy driven incentives for renovation where a building-specific stock description would facilitate better and nuanced information on the impact of policy instruments or financial incentives. In addition to assessing the cost and cost-effectiveness of renovation, socio-economic costs can be calculated to assess affordability [75].

### 2.6 Environmental modelling of building stocks

Life-cycle assessment (LCA) is a commonly used tool to evaluate and assess the environmental impact from buildings. The use of LCA in building-stock modelling has recently become more common [76]. LCA has been used in BSM to investigate the impact of end-of-life stage of building stocks [77], to evaluate the environmental performance of façade renovations in an urban setting [78] and to assess the environmental impact of renovation

measures on the European residential stock [79]. However, the LCA is often limited in terms of impact categories assessed. Most commonly, global warming potential (GWP) is evaluated [80]–[84]. Additional impact categories differ but often consists of indicators such as abiotic depletion potential (ADP) [43], [77] acidification and eutrophication potential (AP, EP) [43], [77], [85], [86], embodied energy (EE) [78], [87], photochemical ozone creation potential (POCP) [43], [77], [85], [86] and ozone depletion potential (ODP) [43], [77], [85], [86] and ozone depletion potential (ODP) [43], [77], [85], [86], [88]. In a few cases, more tailored impact categories such as particulate matter formation (PM10) [89] and embodied water [90] have been used. Furthermore, life-cycle stages generally follow those described in relevant standards [91] but differ as to which are included. Impacts relating to the operational phase is often included [39], [80], [82], [85], [89], [92] as well as impacts from manufacturing of components and materials [78], [85], [92]. Some studies have gone further and include environmental impacts from the building phase [84], [85], [92]. A few examples have been found where maintenance [93] as well as end-of-life [77] stages are included.

## 2.7 Visualization and communication

Results from BSM are commonly visualized on different scales depending on the purpose and stakeholder targeted. In general, visualizations are done on a country [53], regional [50] or city level [94], where models on a city level sometimes highlight a district or neighbourhood [25]. For larger scale visualization for a country or a region, results are commonly visualized in 2D for statistical zones [42] or zones defined by a common urban typology to fit the representative buildings used to describe the stock [95]. On an urban level, results are either visualized in 2D or 3D. 2D visualizations typically represent results for individual buildings or aggregated to areas, where studies of larger cities use areas. 3D, or 2.5D, visualizations of results typically use a district or neighbourhood. If solar energy potential is evaluated, 3D visualizations are done as the higher level of detail is needed [37], [96]. Parameters used in visualizations differ but commonly include energy use or power demand. Other studies use geometric information [42] or typologies [95], [97] in an attempt to draw conclusions by linking such parameters to energy use. To visualize the parametric value, colour coding is often used with a few exceptions where the areas are extruded and the height is used to indicate the parametric value. While many papers mention the ability of these models to provide decision support [98], it is often not explicitly stated which stakeholder the results are aimed at but rather broadly refers to supporting decision making in policy, urban planning and energy planning despite these models targeting specific cities. Furthermore, it is not stated how the spatiotemporal visualization of results are adapted to meet the requirement of the intended stakeholder.

## 2.8 Towards comprehensive building-stock modelling

Current frameworks or classifications for building-stock modelling are based on using representative buildings to model the energy performance of the stock. As data availability increases, the need for representative building descriptions is reduced. A shift from energy use to other indicators for assessing the current and future state of the stock is warranted as energy use in and of itself is rarely of interest. Rather, the environmental impact and especially the greenhouse-gas emissions from the building-stock should be the focal point moving forward

in conjunction with economic modelling to assess the cost-effectiveness of renovation. As such, a wider framework for building-stock modelling is needed. In figure 1, a framework for a more comprehensive building-stock modelling is suggested relating to the spatial scale, the scope, the temporal scale and the assessment. The spatial scale indicates the geographic boundaries and is of importance as modelling the building-stock for an urban area or a country will impose different methodological choices. The scope relates to what part of the stock is being investigated. As has been previously mentioned, modelling requirements are vastly different for residential and non-residential buildings and the object of study will again dictate methodological choices. The temporal scale indicates whether the model considers the current state of the stock, a future state based on fixed trends or a future state based on dynamic evolution of the stock. The assessment defines what is to be studied, divided into environmental, economic and social aspects. The assessment stage also includes presenting and visualizing results appropriate for the intended stakeholders. Energy is not considered as a separate aspect of study, but rather as a prerequisite for a wider assessment. It should be noted that the ability of a building-stock model to account for social aspects is limited. However, there are aspects of affordability and movement patterns relating to renovation that can be studied. In the work presented in this thesis, an urban spatial scale is used to assess the environmental and economic impact of renovation measures applied to the MFB stock in the City of Gothenburg using static and dynamic scenarios. The intended stakeholder is the City of Gothenburg and the municipal housing company.



Figure 2 - Framework for comprehensive building-stock modelling.

## 3. Building-stock information

For this thesis, data were retrieved from the Swedish Mapping, Cadastral and Land Registration Authority, the Swedish Tax Agency, the National Board of Building, Housing and Planning and the City planning office of Gothenburg.

#### 3.1 Data sources

Table 1 shows the building specific databases available for the City of Gothenburg, at what level of aggregation they are available, identifiers used and the information they contain most relevant for this work.

Database/Data owner	Aggregation level	N	Relevant information	Identifier			
The property register (50A)	Building	153 000	Building type, construction year, value year and renovation year	Building ID, Property ID, mid- point coordinates			
(42P)/Swedish Mapping, Cadastral and Land Registration Authority	Property	114 000	Property owner, % owned, income	Property ID			
Gripen/National Board of Building, Housing and Planning	Building	6320	Energy use, HVAC systems, energy performance, heated floor area, number of stories, number of apartments	Building ID			
Cadastral maps/City planning office	Building	178 000	2D, 3D (roofs)	Coordinates			
Swedish Tax Agency	Taxation area	55	Taxation data	Coordinates			

 Table 1 - Building specific databases relevant for analysing the building stock, their level of aggregation and the identifier used to match them.

The Swedish Mapping, Cadastral and Land Registration Authority supplied parts of the property register for the City of Gothenburg. Specifically, they provided the building register (Table 50A) and the register of property owners (Table 42P). The building register contains information on building type, year of construction, year of renovation and value year. The value year is of particular interest as it serves several functions. It is calculated based on the year of construction, year of renovation and the economic extent of previous renovation measures and represents a calculated current state of the building. Hence, it can be used to assess the remaining lifetime of a building and to assess the cost of previous renovation activities. In addition, it is used to calculate the taxation value of a building. Table 2 and Equation 1 describes how the Swedish Tax Office requires a renovation to be registered as a change in value year depending on the cost of the renovation in comparison with new building cost [99]. The Swedish Tax Agency also provided taxation information relating to the location

of buildings where the City of Gothenburg is divided into 55 value areas, which is used together with the value year and the rental income to calculate the taxation value of a building.

·	8				
Renovation cost	Calculation of value year				
Less than 20 % of new	No change in value year				
building cost					
20-70 % of new building cost	The value year is set based on				
	Equation 1				
More than 70 % of new	The value year is set to the				
building cost	year of renovation				

 Table 2 - Calculation of value year based on renovation cost according to the Swedish Tax Office

(Value year-Construction year)	Renovation cost	(1)
Renovation year-construction year	New construction cost	(1)

The National Board of Housing, Building and Planning supplied all EPC for the City of Gothenburg. The Swedish EPC are unique since they not only contain valuable information on characteristics of the buildings such as heating, ventilation and cooling (HVAC) system but also measured energy use for SH, DHW and auxiliary electricity use. However, the Swedish EPC also suffers from some drawbacks. While the energy performance is given separately for SH, DHW and auxiliary electricity use, it is rarely measured as such but rather subdivided by the energy expert issuing the certificate. Similarly, HFA is rarely measured but rather derived based on the living area. In addition, updates are infrequent as the EPC is valid for 10 years unless major changes to the building are done. More information on the Swedish EPC and suggestions on how to overcome issues of deriving the HFA on a stock level have been done in previous work, see Paper I and [100].

From the City planning office, GIS shape files in 2D were provided for the footprints of the buildings, property boundaries as well as outlines of two different levels of areas, so-called base areas which are the lowest level statistical information is presented on and primary areas which are used for administrative purposes. Primary areas typically consist of a dozen base areas.

### 3.2 Combining datasets

The datasets are combined based on the identifiers as follows. The EPC are connected to the property register using the building ID (50A) and the register of property owners is connected to the building register using the property ID. Coordinates is then used to connect these datasets to each individual footprint in the 2D-map of Gothenburg. As not all EPC contain the correct identifier, 5901 of the 6320 EPC are spatially linked to footprints. Similar work has been carried out on a national level using the address to spatially link the databases [101].

# 4. Building-specific stock description and modelling framework

To allow for modelling and assessing energy performance, environmental impact and costeffectiveness of renovating the building-stock, available building-stock information presented in Table 2 is condensed to a description of the building-stock where the technical parameters are characterized. Since all information necessary for energy, environmental and economic modelling are not known directly from the information available, assumptions are made using secondary sources in conjunction with the available information. Further information on the methodology and assumptions made can be found in Paper I, III and IV. In Table 3, the reduced set of data used to describe the building stock of the City of Gothenburg is shown.

Database	Source	Information						
EPC	National board of building, housing and planning	HVAC systems, number of stories, attachme to other buildings, measured energy us number of staircases						
2D-map of Gothenburg	City planning office	Footprint, length and orientation of external walls						
Building register	Swedish mapping, cadastral and land registration authority	Year of construction, year of renovation, value year, owner, rental income						

Table 3 - Reduced set of data used to describe the MFB stock of Gothenburg

## 4.1 Determining the U-value

To estimate a likely U-value, historic building regulations and architectural history books are used. The classification follows the most common method of dividing buildings into different age-type categories. Using an architectural history book [102] containing 32 different MFB types spanning a period of 120 years (1880-2000) together with historic building regulations, average and component specific U-values for an age-type classification have been developed. As not all of the 32 building types listed in the book are unique considering the type but rather construction methods or other architectural features, they are grouped to form overarching categories. In total they are divided into seven different categories and 27 time periods. As all building types the U-value does not change significantly, or at all, over several time periods. In Paper I, average U-values were applied while paper III and IV uses component specific U-values. In paper III, U-values are part of the calibration to better capture the current state of the stock for older buildings were components likely have been replaced.

U-values for the age-type classification are derived from structural drawings provided in [102]. For time periods where there is no knowledge on the structural composition, the U-value is instead taken from the building regulations. National building regulations for Sweden have existed since 1946 with demands connected to the U-value of a building. Due to different climate conditions, regulations have been differentiated by climate zones. All values given in this section refers the City of Gothenburg. From 1946 to 1988, demands on U-values were set at a component level and differentiated between light and heavy constructions, see Table 5. From 1989 to 2006 demand on U-values was instead given as an average U-value for

the entire building and from 2007 and onward the regulations have been based on an average U-value in combination with measured energy use, see table 4. The building code demands an energy performance of a building based on SH, DHW and auxiliary electricity use. As the way demands are set has changed, it is difficult to make a comparison between buildings built before 1989 and those built later as the shape factor, adjacency to other buildings and window to wall area ratio would impact the results.

		domostic ciccultury doct	
Building code	Valid	Average U-value [W/m <sup>2</sup> ,k]	Measured energy use (for buildings with electric heating) [kWh/m <sup>2</sup> ,y]
NR*	1989-1994	0.18 + 0.95 * Aw/Aenv	
BBR* 1-8	1995-2002	0.18 + 0.95 * Aw/Aenv	
BBR 9-11	2003-2006	0.18 + 0.95 * Aw/Aenv	
BBR 12-15	2007-2008	0.5	110 (75)
BBR 16-18	2009-2011	0.5	110 (55)
BBR 19-21	2012-2014	0.4	90 (55)

 

 Table 4 - Demands on average U-value and measured energy use for space heating, domestic hot water and nondomestic electricity use.

\* NR stands for Nybyggnadsregler and BBR stands for Boverkets byggregler, English translation: New construction rules and the national board of building, housing and planning's construction rules. Aw denotes window area and Aenv denotes envelope area.

8	Floc		0.4		0.45		0.4			0.4			0.3		0.2		-040
dish building code from 1946-1988	Wooden roof	construction	0.5		0.45		0.4			0.4			0.2		0.2		Duilding and
	Heavy roof	onstruction	9.		.55		.5			.5			.2		.2		alich turneloti
nents in the Sw	Wood F	c	0.6 0		0.65 0		0.5 0			0.5 0			0.3 0.3		0.3 0.3		
e 5 - Demands on U-value for building compor	Other stone	material	0.8		0.85		0.8			0.8			0.3		0.3		f. Curant P
	ight brick	onstruction	6		95		0			-1			ι,		ω.		J CDNI stands
	y brick L	ruction c	0		0		1			1			0		0		deated and an
Table	Неаvу	constr	1.0		1.05		1.0			1.1			0.3		0.3		Ducent
	Valid		1946-	1950	1951-	1960	1961-	1967		1968-	1975		1976-	1981	1982-	1988	and for
	Building	code (unit)	BABS* 46	(kcal/m², ch)	BABS 50	(kcal/m², ch)	BABS 60	(kcal/m <sup>2</sup> ,	ch)	SBN* 67	(kcal/m <sup>2</sup> ,	ch)	SBN 75	$(W/m^2,k)$	SBN 80	$(W/m^2,k)$	

BABS stands for Byggnadsstadgan and SBN stands for Svensk byggnorm, English translation: Building code and

4.2 Modelling of energy use, environmental impact and cost-effectiveness The building-stock description is used to model the energy performance (Paper I, III and IV), the environmental impact of renovation measures (Paper IV) and the costeffectiveness of deep renovation (Paper III). In paper I and IV, the existing buildingstock energy model ECCABS [20] have been adapted to incorporate spatial information. The model is dynamic, using an adaptive time-step method and treats each building as a single zone. In paper III, an existing building-stock energy models including calibration routine has been used [21]. The model uses a single-zone monthly steady-state method to calculate the energy demand for space heating. For more information on the calibration routine used, see paper III and [21]. On a stock level, both energy models achieve a similar level of accuracy using the same building-stock description while the calibration in the steady-state model provide more accurate results on a building level. The accuracy of the ECCABS model on an urban scale is similar to previous studies on a national scale [34].

The LCA is carried out using the software SimaPro V8.0.5.13 and the database Ecoinvent V3.1. Global data is used in cases where European data is not available. 15 of the ReCIPe [22] V1.12 mid-point categories is used to evaluate the environmental impact of all construction related impact while interior measures for energy efficient lightning and appliances are omitted due to lack of available data. The material use for construction measures is based on a library of common renovation measures on the Swedish market [103], [104]. Several options are considered for measures relating to façade insulation and windows. Emissions from energy use is based on average values for the local energy company for district heating and the Swedish market mix for electricity [105]. Using consequential data is not considered as it is not relevant with regards to local targets being evaluated. The assessment is carried out in accordance with relevant standards [91], [106], using life-cycle stages A1-A3 and B6 with the functional unit being the 5901 MFB used in the assessment.

In order to assess the cost-effectiveness of deep renovation, EAC is used. EAC is calculated for each component and summed up for each building (Eq. 2). The minimum technical lifetime of the component is used as the lifetime of the investment and a discount rate of 4% is used. The EAC does not factor in a change in maintenance cost following renovation.

$$EAC_i = C_i \frac{r}{1 - (1 + r)^{-t_i}}$$
 (Eq. 2)

*EAC:* Equivalent annual costs for a component in [EUR/y]

*Ci:* Investment costs of EEM for a component in [EUR]

r: Discount rate

*ti: lifetime of component i [y]* 

#### 4.3 Assessing changes to the building-stock

To assess changes to the building-stock, three different methods have been used. In paper III, a step change was used in order to evaluate a deep renovation package for the municipal housing company. In paper II, a static scenario was used to assess the implications of current levels of EEM coupled with rate of new construction based on planning documents in order to evaluate the change in energy demand until 2035. In paper IV, two dynamic scenarios were used based on cost-effectiveness of renovation. The two dynamic scenarios have different driving forces as well as two levels of limitations regarding yearly investment cost and maximum share of the stock to be renovated [107]. Scenario 1 assumes building components will be updated at the end of their lifetime, regardless of the cost-effectiveness of the ESM. Scenario 2 considers that ESMs are implemented if they are cost-effective, with the renovation taking place at the end of the lifetime of the building component. The technical lifetime of components is based on EN 15459. Cost-effectiveness of individual measures is evaluated using EAC based on the method described in [108]. Furthermore, renovation packages are prioritized if both individual renovation measure and package is cost-effective. The two scenarios are further divided based on limiting factors based on maximum yearly total HFA being renovated  $(m^2_{HFA}/year)$  and as yearly maximum annual investment capacity per HFA [ $\epsilon/(m^2_{HFA}, year)$ ]. The limitations are not applied on a building level but based on groups of property owners; the municipal housing company, private housing cooperatives and private property owners. Two levels of limiting factors are used. Limitation A is based on average investments in energy efficiency measures by the municipality housing company (7.5 $\in$ /m<sup>2</sup> HFA) [109] and on the national average renovation rate of roughly 1% [17]. Limitation B uses a higher investment capacity of  $10.0 \notin m^2$  HFA and allows for 2.5 % of HFA to be renovated yearly.

The renovation measures applied for assessing deep renovation in the municipal housing stock and evaluating the environmental impact of continuing current renovation practices for the MFB stock are described in detail in paper III and IV respectively.

#### 4.4 Spatial visualisation

Throughout this work, several different spatial visualizations have been used depending on the stakeholder and parameters being evaluated. Yearly energy use has been spatially visualized using different spatial resolutions and units. Results have been presented aggregated to administrative districts such as base areas (935 in Gothenburg) or primary areas (96 in Gothenburg) as well as shown on a building level. Cost-effectiveness have been aggregated to value areas (55 in Gothenburg) or shown on a building level. Environmental impact has not been shown spatially as the aim, and to a certain extent the accuracy, does not warrant such visualizations.

# 5. Results

This chapter starts by presenting results at different levels of aggregation based on the building-stock information that has previously been presented followed by assessing the impact of renovating the building-stock considering energy use, environmental impact and cost-effectiveness. This section contains information related to the entire MFB stock as well as more in-depth information and results related to the municipal housing stock. The effects of renovating the existing stock is a summary based on results in paper II-IV.

## 5.1 Current state of the stock

The City of Gothenburg was founded in 1621 and is the second largest city in Sweden with about 550 000 inhabitants. The urban housing stock of MFB grew outwards until mid-1970s before densification started. The MFB-stock and its spatial distribution is shown in figure 3.



Figure 3 - Spatial distribution of multi-family buildings in the City of Gothenburg shown in red. Figure 4 shows the distribution of HFA in the MFB stock divided by property owner or owner type. The municipality housing company owns about a third (36%) of the stock with another third (33%) being private housing cooperatives. The last third of the stock is owned by private owners, individuals, foundations and others where the last category includes estates and non-profit organisations.



Figure 4 - Percentage of total m<sup>2</sup> of HFA in the stock divided by property owner and owner type.

The technical characteristics for the building-stock of the municipal housing company is shown in Figure 5-9 where the U-value for different components as well as the ventilation system is described. Buildings are grouped per decade and U-vales are given in ranges. The U-values are based on the calibrated building-stock description used in paper III while the ventilation systems are based on data from the EPC. As can be seen, the distribution of U-values of walls and windows within the different age categories is large, especially so for older parts of the building stock which has gone through renovations to varying degree. This supports the view that year of construction is not an ideal indicator of energy performance. The distribution of U-values for the roof follow a similar pattern, although the gap between buildings with well insulated roofs and those without is smaller. This is even more pronounced regarding U-values for the floor, as the relative difference becomes smaller. This is partly due to the insulating effect of the ground. It is also interesting to note that natural ventilation is more or less absent in the stock from the 1960s and onwards and that a central exhaust system is the dominant ventilation system used. The widespread use of a central exhaust system limits the renovation options as a central supply- and exhaust-air system with heat-recovery will require considerable changes to the buildings.



Figure 5 - Distribution of U-value of walls for buildings in the municipal housing stock grouped by year of construction.



Figure 6 - Distribution of U-value of roofs for buildings in the municipal housing stock grouped by year of construction.



Figure 7 - Distribution of U-value of windows for buildings in the municipal housing stock grouped by year of construction.



Figure 8 - Distribution of U-value of floors for buildings in the municipal housing stock grouped by year of construction.



Figure 9 - Distribution of ventilation systems for buildings in the municipal housing stock grouped by year of construction.

Figure 10 shows the measured energy use for SH, DHW and auxiliary electricity use for the MFB stock in Gothenburg. It is divided into age-groups with their associated share of the total HFA. As can be seen, the energy performance is quite even for the stock up until 1980. The sharp decline in energy use in the building-stock occurring during the 1980s can be explained by more stringent demands on U-values being introduced in 1975 as shown in chapter 4. Buildings from 1960-1975 are of particular importance as they constitute the largest part of the stock (42 % of all HFA) and have the highest average energy use (146 kWh/m<sup>2</sup>,y). Unlike buildings from earlier time-periods, these buildings have to a large extent never been renovated and are nearing the end of their service-life, requiring renovation in the coming decade. If substantial reductions in energy demand is to be achieved, EEM needs to be implemented in this part of the stock.



Figure 10 - Measured energy performance for different age-groups of the Gothenburg MFB stock and their respective share of the total stock.

#### 5.2 Renovating the existing stock

This section details the impact of renovating the MFB stock relating to energy use, environmental impact and cost-effectiveness. Environmental impact is assessed for the entire stock while cost-effectiveness is evaluated for the municipal housing stock. Energy use reductions are assessed for both the stock in its entirety as well as the municipal housing stock. The results summarize the findings in paper II-IV.

#### **Energy use**

In Figure 11, energy use reductions for the MFB stock until 2050 are shown for two scenarios and two levels of limiting factors. Scenarios are differentiated based on the renovation logic while limiting factors use two different levels for yearly investment capacity and maximum yearly floor area possible to renovate. For a thorough description of the renovation logic for the scenarios as well as the limiting factors this is provided in chapter 4 and paper IV. Energy use includes SH, DHW and all electricity use (including household electricity use). The total energy use is 3009 GWh/year for the current state of the stock. Changes in energy use over time are small apart from for scenario 1B where a 23 % reduction in yearly energy use by 2050 is achieved. Measures regarding energy efficient lighting and appliances reduce electricity use whilst increasing SH demand. As such, there is a shift from electricity use to district heating, particularly for scenario 2 where energy savings to a large extent is a result of reduced electricity use. As such, while the total yearly energy use is decreased by 0.7 % for scenario 2B and 4.1 % for scenario 2B, the yearly energy use for district heating increases compared to current levels.



Figure 11 - Yearly energy use in the MFB stock until 2050 under different scenarios and limiting factors. Figure 12 shows the energy use for SH and DHW until 2035 for the entire MFB stock based on current implementation rate of EEM as well as accounting for planned new construction. As can be seen, any energy use reduction achieved by renovating the existing stock is offset by increased energy use in new construction. This indicates that current efforts to reduce greenhouse-gas emissions in the existing MFB stock will not be sufficient to reach targets set by the municipality. Hence, to reach local climate goals, substantial improvements on the energy supply side to reduce emissions are needed based on the assumed development scenario.



Figure 12 – Yearly energy use for SH and DHW in the current and future MFB stock.

Figure 13 shows the final energy use for the stock of the municipal housing company after deep renovation grouped by year of construction. The deep renovation applied

results in average final energy use reduction by 51%, to 55 kWh/m<sup>2</sup> year by applying the renovation measure to all buildings. The difference in final energy use after deep renovation does not vary considerably between the different age groups.



Figure 13 – Average final energy use grouped by year of construction for the stock of the municipal housing company after deep renovation.

#### **Environmental impact**

In figure 14, the yearly greenhouse-gas emissions based on energy use reduction in Figure 12 are shown for the MFB stock. Current yearly greenhouse-gas emissions from the MFB stock is 204 ktonCO<sub>2</sub>eq/year. The reduction in yearly emissions follow the same pattern as energy use, but with a larger total reduction. This is due to the shift from electricity use to district heating. This can be exemplified by scenario 1B where the reduction in yearly energy use is 23% while reductions in yearly greenhouse-gas emissions increases to 31% until 2050. It should be noted that no changes to current emission levels are assumed in order to indicate to what extent renovating the building-stock can help meet local climate goals.

![](_page_42_Figure_0.jpeg)

Figure 14 - Yearly greenhouse-gas emissions in the MFB stock until 2050 under different scenarios and limiting factors.

In figure 15, the environmental impact of all construction related measures for scenario 2 until 2050 is given using 15 of the ReCIPe mid-point categories. To indicate tradeoffs, impact categories have been normalised to show the relative impact of individual measures. As results are based on the total uptake of measures across the MFB stock until 2050 it can be used to compare total relative impact but cannot be used to compare individual measures. As opposed to figure 11 and 14, results are only given for scenario 2A as scenario 1 contains a high degree of interior measures relating to lighting and appliances not accounted for in the LCA and there is little difference in the relative impact between scenario 2A and 2B. Installation of PV panels dominates most impact categories, accounting for more than 50% of the environmental impact for nine out of 15 impact categories. Due to the use of package solutions, measures affecting the Uvalue are more comparable. Replacement of windows has the largest relative environmental impact followed by insulation of facades. Roof insulation has a consistently low environmental impact around 3-5% while insulation of floor or basement has an environmental impact ranging from 3-10% with the exception for marine eutrophication where it is responsible for 64% of the environmental impact for scenario 2A. Upgrade of ventilation system with heat recovery have a relatively low environmental impact across all categories, peaking at 10% for metal depletion in scenario 2B.

![](_page_43_Figure_0.jpeg)

Figure 15 - Relative environmental impact of construction related renovation measures implemented until 2050 for 15 mid-point coordinates under scenario 2A.

In figure 16, the cumulative greenhouse-gas emission savings until 2050 for scenario 2A and 2B accounting for changes in energy use and embodied impacts from material use is shown as ktCO<sub>2</sub>eq. As the main difference between the two limiting factors are the total uptake of measures rather than what measures are applied, it serves as an indication of to what extent renovation is beneficial. In both cases, a decreased climate impact is observed with greenhouse-gas emissions saved corresponding to one years' worth of current emission levels. However, the increased number of buildings being renovated in scenario 2B only marginally increases greenhouse-gas emissions saved, from 201 to 211 ktCO<sub>2</sub>eq. Hence, increasing the renovation rate and implementing similar renovation measures for more buildings will have a marginal climate impact and trade-offs relating to other impact categories should be considered.

![](_page_44_Figure_0.jpeg)

Figure 16 – Cumulative climate impact due to energy savings until 2050 and material use for the EEM for scenario 2.

#### **Cost-effectiveness**

In Figure 17, cost-effectiveness is evaluated by showing yearly energy cost savings achieved by deep renovation of the municipal housing stock as a share of EAC. As previously shown, the distribution in U-values is large within the portfolio and as such, cost-effectiveness of deep renovation varies greatly. In addition to a difference in technical characteristics, the compactness plays a major role as well as the fact that certain reinstatement costs scale with number of apartments rather than building size. Annual cost savings account for 21 % of EAC across the municipal housing stock while individual buildings show annual energy cost savings offsetting over 50 % of the annualised cost of deep renovation. As the deep renovation package used is not profitable for any building, additional gains are needed through adapting the renovation measures for individual buildings, achieving lower maintenance cost after renovation or by increasing rent levels.

![](_page_44_Figure_4.jpeg)

Figure 17 – Number of buildings grouped by yearly energy cost savings in relation to equivalent annual cost [%].

Figure 18, yearly energy cost savings as a share of EAC is spatially visualized for a specific value area. Results are presented for both the value area as indicated by the background colour as well as for individual buildings. As can be seen, the cost-effectiveness of deep renovation of individual buildings within the area varies greatly. This highlights one of the pitfalls of aggregating results to areas or zones as the distribution in results for individual buildings can be large.

![](_page_45_Picture_1.jpeg)

Figure 18 - Yearly energy cost savings in relation to equivalent annual cost. Background colour shows average yearly energy cost savings in relation to equivalent annual cost for the value area.

## 6. Discussion and conclusions

The aim of this thesis has been to evaluate the renovation potential of the existing building-stock on an urban level by expanding on methods used in building-stock modelling to enable assessment of environmental impact and cost-effectiveness, to explore the potential for visualizing and communicating results using GIS and in particular to target property owners. In doing so, a methodology for describing building-stocks on a building-by-building basis has been developed and used in assessing environmental impacts of renovating the MFB stock in the City of Gothenburg as well as the cost-effectiveness of deep renovation of the municipal housing stock. The building-specific description in combination with the possibility to validate results on a building level using measured energy use from the EPC allows for results to be presented and communicated at any level of spatial aggregation. This has enabled exploration of possible ways to visualize and communicate results to different stakeholders involved in urban transformation. As such, the research questions posed have been answered and the findings are summarized below.

RQ1 – What is the potential for a building-stock description where each building is treated individually?

As shown in paper I, III and IV it is possible to create a building-stock description where each building is treated individually. The methodology has expanded from paper I where average U-values for buildings was used to a component level differentiation in paper IV which was needed in order to assess renovation options. In paper III, the issue of calibrating the description on a building level was dealt with, resulting in increased accuracy on a building level. By creating a building-specific stock description, a wider range of stakeholders could be targeted and a more nuanced understanding of the current state of the building stock achieved as well as a differentiated view on the impact of renovation measures.

RQ2 – How can GIS be used with a building-specific stock description to visualize and communicate results for a wider range of stakeholders, including property owners?

In paper II, different spatiotemporal resolutions were explored in visualizing results to communicate the information regarding the current state and possible future development of the MFB stock to specific stakeholders. In paper III and IV, spatial visualizations and aggregation of results where done with specific stakeholders in mind. Information relating to energy use, environmental performance and cost-effectiveness have been visualized using several scales and metrics to target specific stakeholders. The visualizations used here should be seen as a first step and further work is needed in assessing what metrics and level of aggregation is suitable depending on the intended stakeholder. Additionally, for many stakeholders the housing stock is only of partial interest as part of a larger system.

RQ3 – How can the financial viability of (deep) renovation be assessed across a building portfolio?

As shown in paper III, assessing and differentiating cost-effectiveness of deep renovation within a building portfolio using EAC and change in assessed building value is possible. Deep renovation is certainly not suitable for all buildings in the portfolio, neither from an energy nor cost-effectiveness point of view. Rather, the results serve as a first step to prioritize buildings and areas or neighbourhoods to differentiate where deep renovation is suitable within a large building portfolio. Further detailed assessment of buildings suitable for deep renovation is needed as well as tailoring of the renovation measures to each building. Ideally, the financial viability should be assessed using the yield by incorporating maintenance costs.

RQ4 – What are the environmental impacts of current renovation trends in an urban building-stock?

In paper IV, the environmental impact of renovating the existing MFB stock in the City of Gothenburg until 2050 using business-as-usual scenarios has been done. The use of business-as-usual scenarios highlights the limited impact current renovation rate and measures will have and acts as a counterbalance to estimations of technical potential. As the measures used are conservative, a more aggressive renovation strategy could be investigated as the marginal costs are likely to be small. However, renovation measures with a larger energy reduction potential may prove to be inefficient or detrimental in reducing greenhouse-gas emissions due to the relatively low emissions from district heating while negatively affecting other environmental impact categories. In addition, the possibility of including further life-cycle stages in the LCA as well as assessing local impacts using national data could be investigated. Depending on the intended stakeholder, using select end-point indicators or tailored weightings should be considered.

This thesis with appended papers has shown that available data sources can be used to describe the characteristics of the stock on a building level while considering the location and context. Building attributes such as year of construction, value year and owner from the Swedish Property register, spatial data provided by the City planning office of Gothenburg as well as EPC from the national board of housing, building and planning can provide detailed and valuable information to policy makers, urban planners and property owners. Being able to target property owners is especially important as this is something that has been largely missing from previous research. In this work, the potential for renovating the MFB stock in the City of Gothenburg has been assessed. The need for renovation provides an opportunity for achieving considerable reductions in energy demand and associated emissions of green-house gases. This work gives an overview of the existing situation and provides a thorough assessment of the potentials and impacts associated with renovating the existing MFB stock. However, the issue of profitability in renovation persists and the potential reduction in greenhouse-gases is limited.

#### Limitations

There are still limitations regarding understanding the current state of the stock. Information regarding past renovation activities is limited and the level of maintenance is unknown. Furthermore, the databases used can be updated, especially the EPC as many of them are a decade old. The use of a 3D-model of the City of Gothenburg could be used to better assess the geometry of buildings. In addition to technical characteristics, the user-behavior is simplified and in order to improve the accuracy of calculated energy use, load curves can be used.

The main drawback of a building-specific bottom-up model is the extensive need for granular data. As such, applying such a description on a national or European level is difficult. As an alternative to using representative buildings for national or pan-national stock models, a hybrid approach can be used. These approaches can achieve a similar level of detail as a building-specific stock description at the cost of spatial granularity. By creating artificial distributions from representative buildings, a synthetic stock description can be used [21]. Another example of a hybrid method is the Canadian hybrid residential end-use energy and greenhouse gas Emissions model [110] using a large sample of buildings coupled with a neural network.

#### Visualization and communication with stakeholders

The spatial visualization of renovation activities related to individual buildings can be used to support communication with urban planners, policy makers and property owners regarding coming renovation needs in an urban context. It is also clear from this work that while the renovation need will be large in the near future, information at aggregate levels need to be used with care when suggesting areas where measures are to be prioritized. Further work is needed to assess what information different stakeholders involved in the transformation of the urban building stock needs and how it should be presented and visualized.

While information relating to the building stock constitutes partial information to individual stakeholders, it opens the possibility to facilitate communication and joint understanding between the different stakeholders. While spatial representation is an important tool in promoting joint understanding, it is not always relevant to all stakeholders. For property owners, spatially visualizing energy related information only make sense for portfolios of sufficient size. However, to spatially visualize results with this level of detail is not always needed and the spatial aspect may not always be warranted. GIS may still be used for practical reasons regarding data acquisition, merging and analysis. One of the main benefits of a bottom-up modelling approach is the ability to aggregate results to different scales depending on the use case. Care should be taken not to visually represent data at a scale at which the methods used to derive it is not accurate. As such, the practice of visualizing results on a building level when distributing aggregated data or when results are based on archetypes should be avoided.

#### A building-specific description for a heterogenous building stock

The effect of renovation on energy use, environmental impact and cost-effectiveness varies greatly within the stock. While the average effect of renovation based on current practices provide marginal energy use reductions and associated reductions in greenhouse-gas emissions are small, individual buildings can be identified within the stock where the potential is larger. Similarly, the cost-effectiveness of deep renovation shows a wide distribution across the municipal housing portfolio. As such, going forward it is suggested that key buildings within the building stock suitable for large scale renovation are identified. All buildings are unique which is why a building-specific approach in describing the building stock is required. While there are technical similarities between some buildings, aspects such as current state, location, owner and tenants ensures that each building deserves to be treated individually when assessing potential renovation measures.

In the Swedish context, the approach towards renovation have shifted in the past years. In the past, deep renovation was argued for and several methodologies were developed to support such endeavours. Recently, the pendulum has swung and property owners such as the municipal housing company in Gothenburg now argues for "gentle renovation". The results of this thesis show that there is a need for a more differentiated view on renovation. Applying a building-specific stock description that can capture the distribution within the stock is thus warranted.

### 7. Future research

The methodology used in this thesis to develop and apply a building-specific stock description can be further developed. With digitalization, more information is bound to become accessible which would allow for creating a digital twin [111], [112] of a city that can be used to in detail describe the current state and evaluate changes in the built environment.

To better assess the impact of renovating the stock, methodological advancements are necessary. While steps have been taken in this work to better describe the current state of the stock, there are still room for improvement. By incorporating maintenance plans and inspection reports, the current state of the stock would be better understood. On the modelling side, energy modelling should better capture the user behaviour in order to allow for better understanding of load curves and what measures and interventions affect them. This becomes increasingly important as greenhouse-gas emissions are tied to peak demand. Similarly, to be able to better assess the environmental impact of renovation measures, local datasets should be developed for LCA and local environmental impacts estimated and spatially visualized. This is especially important in developed countries where renovation measures may end up moving local emissions to a global scale. Such trade-offs should be highlighted to better support policy makers. In parallel to increased spatial resolution of environmental impacts, the temporal resolution should be increased as emissions from energy use vary over the year and time of day. Furthermore, evaluating cost-effectiveness of renovation measures should also include maintenance costs [113] to allow for evaluating the yield to better align with common decision practices for property owners. For many of the stakeholders targeted, the MFB stock is only of partial interest. To better assess the building-stock as a whole, describing and modelling the energy use of non-residential buildings needs further attention. Lastly, the use of a detailed description for a bottom-up engineering model for evaluating policy measures and subsidies regarding renovation should be investigated.

#### 8. References

- [1] BPIE, Europe's Buildings Under the Microscope. 2011.
- [2] IPCC, 'Climate Change 2014: Mitigation of Climate Change: Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on', *Cambridge Univ. Press.* ..., p. 1132 pp., 2014.
- [3] E. Recast, 'Directive 2010/31/EU of the European Parliament and of the Council of 19 May 2010 on the energy performance of buildings (recast)', *Off. J. Eur. Union*, vol. 18, no. 06, p. 2010, 2010.
- [4] E. N. Directive, '91/EC of the European Parliament and of the Council of 16 December 2002 on the energy performance of buildings', Off. J. Eur. Communities, vol. 4, no. 2003, p. L1, 2002.
- [5] SOU, 'Nationellt program för energieffektivisering och energismart byggande, prop. 2005/06:145', 2006.
- [6] The Swedish Government, Swedish Government Bill 2008/09:163, A cohesive energy and climate policy: Energy. 2009.
- [7] City of Gothenburg, 'Climate programme for Gothenburg', 2014.
- [8] U. Sbci, 'Buildings and climate change: summary for decision-makers', *United Nations Environ. Program. Sustain. Build. Clim. Initiat. Paris*, pp. 1–62, 2009.
- [9] Boverket, 'Flyttmönster till följd av omfattande renovering', Karlskrona, 2014.
- [10] SCB, 'The Swedish housing stock', 2019.
- [11] M. Hedenmo and F. von Platen, 'Bostadspolitiken: Svensk politik för boende, byggande och planering under 130 år [Elektronisk version]', *Karlskrona Boverk.*, 2007.
- [12] Industrifakta, 'Behov och prioriteringar i rekordårens flerbostadshus', 2011.
- [13] P. Lindqvist, J. Jönsson, and T. Ekvall, 'Miljonprogrammet-Förutsättningar och möjligheter', *Prognoscentret*, 2013.
- [14] P. Lindqvist, J. Jönsson, and T. Ekvall, 'Renoveringsbehov i Miljonprogrammet – Uppföljningsstudie september 2018', *Prognoscentret*, 2018.
- [15] Swedish Energy Agency, 'Energy statistics for multi-dwelling buildings 2017', 2018.
- [16] Swedish Energy Agency, 'Energy statistics for multi-dwelling buildings in 2016', 2016.
- [17] Boverket, 'Underlag till den andra nationella strategin för energieffektiviserande renovering. Ett samarbete mellan Boverket och Energimyndigheten', Karlskrona, 2016.
- [18] K. Högdal, 'Halvera Mera-Slutrapport', Stock. BeBo (Energy Agency Purch. Gr. energy Effic. Apartm. Build., 2013.
- [19] Å. Wahlström and M.-L. Maripuu, 'A method presenting economic rationales for major reduction of energy use in non-residential buildings', 2017.
- [20] É. Mata, A. S. Kalagasidis, and F. Johnsson, 'A modelling strategy for energy, carbon, and cost assessments of building stocks', *Energy Build.*, vol. 56, pp. 100–

108, 2013.

- [21] C. Nägeli, C. Camarasa, M. Jakob, G. Catenazzi, and Y. Ostermeyer, 'Synthetic Building Stocks as a Way to Assess the Energy Demand and Greenhouse Gas Emissions of National Building Stocks', *Energy Build.*, 2018.
- [22] R. Zelm, 'ReCiPe 2008: a life cycle impact assessment method which comprises harmonised category indicators at the midpoint and the endpoint level', *Den Haag, NetherlandsGuinee JB, Gorree M, Heijungs R, Huppes G, Kleijn R, Kon. A, van Oers L, Sleeswijk AW, Suh S, Udo Haes HA, Bruijn H, van Duin R, Huijbregts MAJ Life cycle Assess. an Oper. Guid. to ISO Stand. eco*, vol. 7, p. 445460, 2009.
- [23] S. K. Firth, K. J. Lomas, and A. J. Wright, 'Targeting household energyefficiency measures using sensitivity analysis', *Build. Res. Inf.*, vol. 38, no. 1, pp. 24–41, 2010.
- [24] F. Ascione, R. F. De Masi, F. de Rossi, R. Fistola, M. Sasso, and G. P. Vanoli, 'Analysis and diagnosis of the energy performance of buildings and districts: Methodology, validation and development of Urban Energy Maps', *Cities*, vol. 35, pp. 270–283, 2013.
- [25] G. Dall'o', A. Galante, and M. Torri, 'A methodology for the energy performance classification of residential building stock on an urban scale', *Energy Build.*, vol. 48, pp. 211–219, 2012.
- [26] K. Fabbri, M. Zuppiroli, and K. Ambrogio, 'Heritage buildings and energy performance: Mapping with GIS tools', *Energy Build.*, vol. 48, pp. 137–145, 2012.
- [27] S. Heiple and D. J. Sailor, 'Using building energy simulation and geospatial modeling techniques to determine high resolution building sector energy consumption profiles', *Energy Build.*, vol. 40, no. 8, pp. 1426–1436, 2008.
- [28] J. Kragh and K. B. Wittchen, 'Development of two Danish building typologies for residential buildings', *Energy Build.*, vol. 68, no. PARTA, pp. 79–86, 2014.
- [29] L. G. Swan and V. I. Ugursal, 'Modeling of end-use energy consumption in the residential sector: A review of modeling techniques', *Renewable and Sustainable Energy Reviews*, vol. 13, no. 8. pp. 1819–1835, 2009.
- [30] M. Kavgic, A. Mavrogianni, D. Mumovic, A. Summerfield, Z. Stevanovic, and M. Djurovic-Petrovic, 'A review of bottom-up building stock models for energy consumption in the residential sector', *Build. Environ.*, vol. 45, no. 7, pp. 1683– 1697, 2010.
- [31] C. F. Reinhart and C. Cerezo Davila, 'Urban building energy modeling A review of a nascent field', *Build. Environ.*, vol. 97, pp. 196–202, 2016.
- [32] W. Li *et al.*, 'Modeling urban building energy use: A review of modeling approaches and procedures', *Energy*, vol. 141, pp. 2445–2457, 2017.
- [33] I. Sartori, B. J. Wachenfeldt, and A. G. Hestnes, 'Energy demand in the Norwegian building stock: Scenarios on potential reduction', *Energy Policy*, vol. 37, no. 5, pp. 1614–1627, 2009.
- [34] É. Mata, A. Sasic Kalagasidis, and F. Johnsson, 'Building-stock aggregation through archetype buildings: France, Germany, Spain and the UK', *Build*.

*Environ.*, vol. 81, pp. 270–282, 2014.

- [35] A. Mastrucci, O. Baume, F. Stazi, and U. Leopold, 'Estimating energy savings for the residential building stock of an entire city: A GIS-based statistical downscaling approach applied to Rotterdam', *Energy Build.*, vol. 75, pp. 358– 367, 2014.
- [36] R. McKenna, E. Merkel, D. Fehrenbach, S. Mehne, and W. Fichtner, 'Energy efficiency in the German residential sector: A bottom-up building-stock-modelbased analysis in the context of energy-political targets', *Build. Environ.*, vol. 62, pp. 77–88, 2013.
- [37] I. Theodoridou, M. Karteris, G. Mallinis, A. M. Papadopoulos, and M. Hegger, 'Assessment of retrofitting measures and solar systems' potential in urban areas using Geographical Information Systems: Application to a Mediterranean city', *Renew. Sustain. Energy Rev.*, vol. 16, no. 8, pp. 6239–6261, 2012.
- [38] M. Jakob, H. Wallbaum, G. Catenazzi, G. Martius, C. Nägeli, and B. Sunarjo, 'A Methodology for spatial modelling of energy and resource use of buildings in urbanized areas', *Cisbat 2013*, pp. 1047–1052, 2013.
- [39] N. Heeren, M. Jakob, G. Martius, N. Gross, and H. Wallbaum, 'A component based bottom-up building stock model for comprehensive environmental impact assessment and target control', *Renew. Sustain. Energy Rev.*, vol. 20, pp. 45–56, 2013.
- [40] A. Mavrogianni, M. Davies, M. Kolokotroni, and I. Hamilton, 'A GIS-based bottom-up space heating demand model of the London domestic stock', in *Proceedings 11th International IBPSA Conference, Building Simulation*, 2009.
- [41] M. Jakob, H. Wallbaum, G. Catenazzi, G. Martius, C. Nägeli, and B. Sunarjo, 'Spatial building stock modelling to assess energy-efficiency and renewable energy in an urban context', in CISBAT 2013-International Conference-Cleantech for smart cities & buildings from nano to urban scale, 2013.
- [42] P. Steadman, I. Hamilton, and S. Evans, 'Energy and urban built form: An empirical and statistical approach', *Build. Res. Inf.*, vol. 42, no. 1, pp. 17–31, 2014.
- [43] A. Mastrucci, E. Popovici, A. Marvuglia, L. De Sousa, E. Benetto, and U. Leopold, 'GIS-based Life Cycle Assessment of urban building stocks retrofittinga bottom-up framework applied to Luxembourg', *Proc. EnviroInfo ICT Sustain.* 2015, no. EnviroInfo, pp. 47–56, 2015.
- [44] M. Österbring, É. Mata, L. Thuvander, M. Mangold, F. Johnsson, and H. Wallbaum, 'A differentiated description of building-stocks for a georeferenced urban bottom-up building-stock model', *Energy Build.*, vol. 120, pp. 78–84, 2016.
- [45] B. Eriksson, *Energisparpotentialer i bostadsbeståndet: värmebalansmodell*. Statens Institut för byggnadsforskning, 1993.
- [46] M. Aksoezen, M. Daniel, U. Hassler, and N. Kohler, 'Building age as an indicator for energy consumption', *Energy Build.*, vol. 87, pp. 74–86, 2015.
- [47] M. Gangolells, M. Casals, N. Forcada, M. MacArulla, and E. Cuerva, 'Energy mapping of existing building stock in Spain', *J. Clean. Prod.*, vol. 112, no. June 2013, pp. 3895–3904, 2016.

- [48] J. A. Fonseca and A. Schlueter, 'Integrated model for characterization of spatiotemporal building energy consumption patterns in neighborhoods and city districts', *Appl. Energy*, vol. 142, pp. 247–265, 2015.
- [49] P. Caputo, G. Costa, and S. Ferrari, 'A supporting method for defining energy strategies in the building sector at urban scale', *Energy Policy*, vol. 55, pp. 261– 270, 2013.
- [50] S. Nielsen and B. Möller, 'GIS based analysis of future district heating potential in Denmark', *Energy*, vol. 57, pp. 458–468, 2013.
- [51] R. Nouvel, A. Mastrucci, U. Leopold, O. Baume, V. Coors, and U. Eicker, 'Combining GIS-based statistical and engineering urban heat consumption models: Towards a new framework for multi-scale policy support', *Energy Build.*, vol. 107, pp. 204–212, 2015.
- [52] J. A. Fonseca, T. A. Nguyen, A. Schlueter, and F. Marechal, 'City Energy Analyst (CEA): Integrated framework for analysis and optimization of building energy systems in neighborhoods and city districts', *Energy Build.*, vol. 113, pp. 202–226, 2016.
- [53] S. Petrovic and K. Karlsson, 'Model for Determining Geographical Distribution of Heat Saving Potentials in Danish Building Stock', *ISPRS Int. J. Geo-Information*, vol. 3, no. 1, pp. 143–165, 2014.
- [54] C. Delmastro, G. Mutani, and L. Schranz, 'The evaluation of buildings energy consumption and the optimization of district heating networks: a GIS-based model', *Int. J. Energy Environ. Eng.*, vol. 7, no. 3, pp. 343–351, 2016.
- [55] M. Mangold, M. österbring, C. Overland, T. Johansson, and H. Wallbaum, 'Building ownership, renovation investments, and energy performance-A study of multi-family dwellings in Gothenburg', *Sustain.*, vol. 10, no. 5, 2018.
- [56] D. Majcen, L. Itard, and H. Visscher, 'Actual heating energy savings in thermally renovated Dutch dwellings', *Energy Policy*, vol. 97, pp. 82–92, 2016.
- [57] P. van den Brom, A. Meijer, and H. Visscher, 'Actual energy saving effects of thermal renovations in dwellings—longitudinal data analysis including building and occupant characteristics', *Energy Build.*, vol. 182, pp. 251–263, 2019.
- [58] F. Filippidou, N. Nieboer, and H. Visscher, 'Are we moving fast enough? The energy renovation rate of the Dutch non-profit housing using the national energy labelling database', *Energy Policy*, vol. 109, no. July, pp. 488–498, 2017.
- [59] G. Happle, J. A. Fonseca, and A. Schlueter, 'A review on occupant behavior in urban building energy models', *Energy Build.*, vol. 174, pp. 276–292, 2018.
- [60] R. J. Beckman, K. A. Baggerly, and M. D. McKay, 'Creating synthetic baseline populations', *Transp. Res. Part A Policy Pract.*, vol. 30, no. 6, pp. 415–429, 1996.
- [61] A. T. Booth, R. Choudhary, and D. J. Spiegelhalter, 'Handling uncertainty in housing stock models', *Build. Environ.*, vol. 48, no. 1, pp. 35–47, 2012.
- [62] H. Tommerup and S. Svendsen, 'Energy savings in Danish residential building stock', *Energy Build.*, vol. 38, no. 6, pp. 618–626, 2006.
- [63] G. Semprini, R. Gulli, and A. Ferrante, 'Deep regeneration vs shallow renovation to achieve nearly Zero Energy in existing buildings: Energy saving and economic

impact of design solutions in the housing stock of Bologna', *Energy Build.*, vol. 156, pp. 327–342, 2017.

- [64] J. J. Mcarthur and C. G. H. Jofeh, 'Portfolio retrofit evaluation: A methodology for optimizing a large number of building retrofits to achieve triple-bottom-line objectives', *Sustain. Cities Soc.*, vol. 27, pp. 263–274, 2016.
- [65] R. and T. Mason-Jones D.R., *Exploring the relationship between the sustainability of construction and market value.* 1981.
- [66] N. Miller, J. Spivey, and A. Florance, 'Does Green Pay Off?', J. Real Estate Portf. Manag., vol. 14, no. 4, pp. 385–400, 2008.
- [67] A. Audenaert, L. De Boeck, and K. Roelants, 'Economic analysis of the profitability of energy-saving architectural measures for the achievement of the EPB-standard', *Energy*, vol. 35, pp. 2965–2971, 2010.
- [68] E. Asadi, M. Gameiro Da Silva, C. H. Antunes, and L. Dias, 'Multi-objective optimization for building retrofit strategies: A model and an application', *Energy Build.*, vol. 44, pp. 81–87, 2011.
- [69] E. Kazimieras Zavadskas, A. R. Kaklauskas, and S. Raslanas, 'Evaluation of investments into housing renovation', *Int. J. Strateg. Prop. Manag.*, vol. 83, no. 8, pp. 177–190, 2004.
- [70] E. Pikas, J. Kurnitski, R. Liias, and M. Thalfeldt, 'Quantification of economic benefits of renovation of apartment buildings as a basis for cost optimal 2030 energy efficiency strategies', *Energy Build.*, vol. 86, pp. 151–160, 2015.
- [71] S. Stanley, R. C. Lyons, and S. Lyons, 'The price effect of building energy ratings in the Dublin residential market', *Energy Effic.*, 2053.
- [72] M. Hyland, R. C. Lyons, and S. Lyons, 'The value of domestic building energy efficiency evidence from Ireland', *Energy Econ.*, vol. 40, pp. 943–952, 2013.
- [73] F. Fuerst, P. Mcallister, A. Nanda, and P. Wyatt, 'Energy performance ratings and house prices in Wales: An empirical study', 2016.
- [74] R. P. Pascuas, G. Paoletti, and R. Lollini, 'Impact and reliability of EPCs in the real estate market', *Energy Procedia*, vol. 140, pp. 102–114, Dec. 2017.
- [75] M. Mangold, M. Österbring, H. Wallbaum, L. Thuvander, and P. Femenias, 'Socio-economic impact of renovation and energy retrofitting of the Gothenburg building stock', *Energy Build.*, vol. 123, pp. 41–49, 2016.
- [76] A. Mastrucci, A. Marvuglia, U. Leopold, and E. Benetto, 'Life Cycle Assessment of building stocks from urban to transnational scales: A review', *Renew. Sustain. Energy Rev.*, vol. 74, no. September 2016, pp. 316–332, 2017.
- [77] A. Mastrucci, A. Marvuglia, E. Popovici, U. Leopold, and E. Benetto, 'Geospatial characterization of building material stocks for the life cycle assessment of end-of-life scenarios at the urban scale', *Resour. Conserv. Recycl.*, vol. 123, pp. 54–66, 2017.
- [78] S. García-Pérez, J. Sierra-Pérez, and J. Boschmonart-Rives, 'Environmental assessment at the urban level combining LCA-GIS methodologies: A case study of energy retrofits in the Barcelona metropolitan area', *Build. Environ.*, vol. 134, no. January, pp. 191–204, 2018.

- [79] K. Allacker, V. Castellani, G. Baldinelli, F. Bianchi, C. Baldassarri, and S. Sala, 'Energy simulation and LCA for macro-scale analysis of eco-innovations in the housing stock', *Int. J. Life Cycle Assess.*, 2018.
- [80] N. W. O. Brown, S. Olsson, and T. Malmqvist, 'Embodied greenhouse gas emissions from refurbishment of residential building stock to achieve a 50% operational energy reduction', *Build. Environ.*, vol. 79, pp. 46–56, 2014.
- [81] P. Chastas, T. Theodosiou, K. J. Kontoleon, and D. Bikas, 'Normalising and assessing carbon emissions in the building sector: A review on the embodied CO2emissions of residential buildings', *Build. Environ.*, vol. 130, no. August 2017, pp. 212–226, 2018.
- [82] J. E. Anderson, G. Wulfhorst, and W. Lang, 'Expanding the use of life-cycle assessment to capture induced impacts in the built environment', *Build. Environ.*, vol. 94, pp. 403–416, 2015.
- [83] Q. Wang, R. Laurenti, and S. Holmberg, 'A novel hybrid methodology to evaluate sustainable retrofitting in existing Swedish residential buildings', *Sustain. Cities Soc.*, vol. 16, pp. 24–38, 2015.
- [84] S. Pauliuk, K. Sjöstrand, and D. B. Muller, 'Transforming the Norwegian dwelling stock to reach the 2 degrees celsius climate target: Combining material flow analysis and life cycle assessment techniques', *J. Ind. Ecol.*, vol. 17, no. 4, pp. 542–554, 2013.
- [85] R. Moschetti, L. Mazzarella, and N. Nord, 'An overall methodology to define reference values for building sustainability parameters', *Energy Build.*, vol. 88, pp. 413–427, 2015.
- [86] F. Nemry *et al.*, 'Options to reduce the environmental impacts of residential buildings in the European Union-Potential and costs', *Energy Build.*, vol. 42, no. 7, pp. 976–984, 2010.
- [87] J. L. Reyna and M. V. Chester, 'The Growth of Urban Building Stock: Unintended Lock-in and Embedded Environmental Effects', *J. Ind. Ecol.*, vol. 19, no. 4, pp. 524–537, 2015.
- [88] N. Kohler and W. Yang, 'Long-term management of building stocks', *Build. Res. Inf.*, vol. 35, no. 4, pp. 351–362, 2007.
- [89] D. Saner, C. Vadenbo, B. Steubing, and S. Hellweg, 'Regionalized LCA-based optimization of building energy supply: Method and case study for a swiss municipality', *Environ. Sci. Technol.*, vol. 48, no. 13, pp. 7651–7659, 2014.
- [90] A. Stephan and A. Athanassiadis, 'Corrigendum to "Quantifying and mapping embodied environmental requirements of urban building stocks" (Building and Environment (2017) 114 (187–202) (S0360132316304747) (10.1016/j.buildenv.2016.11.043))', Build. Environ., vol. 121, pp. 291–292, 2017.
- [91] E. C. for S. (CEN), 'EN 15978: 2011–Sustainability of construction works– Assessment of environmental performance of buildings–Calculation method'. 2011.
- [92] A. Stephan, R. H. Crawford, and K. de Myttenaere, 'Multi-scale life cycle energy analysis of a low-density suburban neighbourhood in Melbourne, Australia', *Build. Environ.*, vol. 68, pp. 35–49, 2013.

- [93] M. Almeida, M. Ferreira, and R. Barbosa, 'Relevance of Embodied Energy and Carbon Emissions on Assessing Cost Effectiveness in Building Renovation— Contribution from the Analysis of Case Studies in Six European Countries', *Buildings*, vol. 8, no. 8, p. 103, 2018.
- [94] J. Mikkola and P. D. Lund, 'Models for generating place and time dependent urban energy demand profiles', *Appl. Energy*, vol. 130, pp. 256–264, 2014.
- [95] J. Pittam, P. D. O'Sullivan, and G. O'Sullivan, 'Stock aggregation model and virtual archetype for large scale retrofit modelling of local authority housing in Ireland', *Energy Procedia*, vol. 62, no. 0, pp. 704–713, 2014.
- [96] J. A. Jakubiec and C. F. Reinhart, 'A method for predicting city-wide electricity gains from photovoltaic panels based on LiDAR and GIS data combined with hourly Daysim simulations', *Sol. Energy*, vol. 93, pp. 127–143, 2013.
- [97] H. Tanikawa and S. Hashimoto, 'Urban stock over time: Spatial material stock analysis using 4d-GIS', *Build. Res. Inf.*, vol. 37, no. 5–6, pp. 483–502, 2009.
- [98] M. Österbring, L. Thuvander, É. Mata, and H. Wallbaum, 'Stakeholder Specific Multi-Scale Spatial Representation of Urban Building-Stocks', *ISPRS Int. J. Geo-Information*, vol. 7, no. 5, p. 173, 2018.
- [99] Swedish tax agency, 'SKV A 2015:14', 2015.
- [100] M. Mangold, M. Österbring, and H. Wallbaum, 'Handling data uncertainties when using Swedish energy performance certificate data to describe energy usage in the building stock', *Energy Build.*, vol. 102, pp. 328–336, 2015.
- [101] T. Johansson, T. Olofsson, and M. Mangold, 'Development of an energy atlas for renovation of the multifamily building stock in Sweden', *Appl. Energy*, vol. 203, pp. 723–736, 2017.
- [102] C. Björk, P. Kallstenius, and L. Reppen, Så byggdes husen 1880-2000: arkitektur, konstruktion och material i våra flerbostadshus under 120 år. Svensk Byggtjänst, 2013.
- [103] Wikells, Sektionsfakta-VSS 17/18 Teknisk-ekonomisk sammanställning av VVSbyggdelar (Technical and financial compilation of HVAC parts). Växjö, Sweden: Wikells Byggberäkningar AB, 2017.
- [104] Wikells, Sektionsfakta-ROT 17/18 Teknisk-ekonomisk sammanställning av ROTbyggdelar (Technical and financial compilation of building parts). Växjö, Sweden: Wikells Byggberäkningar AB, 2017.
- [105] F. Martinsson, J. Gode, J. Arnell, and J. Höglund, 'Emissionsfaktorer för nordisk elproduktionsmix', *IVL Rapp. B*, vol. 2118, 2012.
- [106] E. C. for S. (CEN), 'EN 15643–2: 2011–Sustainability of construction works– Sustainability assessment of buildings–Part 2: Framework for the assessment of environmental performance'. 2011.
- [107] É. Mata, J. Wanemark, M. Österbring, L. Thuvander, and H. Wallbaum, 'Decision-Making in Building Retrofitting: Lessons from Dynamic Modelling of Scenarios for Gothenburg City', in 5th European Conference on Behaviour and Energy Efficiency, 2018.
- [108] É. Mata and A. S. Kalagasidis, 'Cost-effective retrofitting of Swedish residential buildings : effects of energy price developments and discount rates', *Energy*

*Effic.*, pp. 223–237, 2015.

- [109] Framtiden AB, 'Årsrapport 2017', Göteborg, 2018.
- [110] L. G. Swan, V. I. Ugursal, and I. Beausoleil-Morrison, 'Occupant related household energy consumption in Canada: Estimation using a bottom-up neural-network technique', *Energy Build.*, vol. 43, no. 2–3, pp. 326–337, 2011.
- [111] S. Boschert and R. Rosen, 'Digital twin—the simulation aspect', in *Mechatronic Futures*, Springer, 2016, pp. 59–74.
- [112] F. Tao, J. Cheng, Q. Qi, M. Zhang, H. Zhang, and F. Sui, 'Digital twin-driven product design, manufacturing and service with big data', *Int. J. Adv. Manuf. Technol.*, vol. 94, no. 9–12, pp. 3563–3576, 2018.
- [113] A. Farahani, H. Wallbaum, and J.-O. Dalenbäck, 'The importance of life-cycle based planning in maintenance and energy renovation of multifamily buildings', *Sustain. Cities Soc.*, vol. 44, pp. 715–725, 2019.